Author's response to reviews

Title: Concurrent pulmonary zygomycosis and Mycobacterium tuberculosis infection: a case report

Authors:

Tejal Patel (drtejalpatel@yahoo.com)
Ian J Clifton (ian.clifton@leedsth.nhs.uk)
Jack A Kastelik (j.a.kastelik@hull.ac.uk)
Daniel G Peckham (daniel.peckham@btinternet.com)

Version: 8 Date: 2 May 2007

Author's response to reviews:

Dear Prof Kidd,

Re: Concurrent pulmonary zygomycosis and Mycobacterium tuberculosis infection: a case report

We have made the following adjustments to the manuscript.

Removed "Dr" from the author list on the title page.

Removed "Figure 1" from the figure.

Yours sincerely,

Dr Ian Clifton